Dear Editors and Reviewers:

Thank you for your favorite consideration and the reviewer’s insightful comments of our manuscript (ID: 73207). Those comments are all valuable and very helpful for revising and improving our paper. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the reviewer’s comments:

Reviewer 1:
1. Some minor spelling errors should be corrected before final acceptance.
Response: Thank you very much for pointing out this problem. We have corrected all the minor spelling errors.

Reviewer 3:
1. Some minor language polishing should be revised.
Response: Thank you very much for pointing out this problem. We have corrected all the minor language polishing.
2. The limit of the study should be listed and discussed.
Response: Thank you very much for your suggestion. We have listed and discussed the limitations of this study. The modifications are as follows.

“The limitation of this study is that it is a small sample and single center study. Based on the results of this study, we believe that biofeedback therapy and Baduanjin training can improve the quality of life and gastrointestinal hormone levels in patients with colorectal cancer, and the combined effect is superior to monotherapy. As an economic and effective means of clinical and family intervention, it is worthy of promotion and application in clinical and community-based family rehabilitation.”

3. Please move the tables to the end of the text.
Response: Thank you very much for your suggestion. We have moved the tables to the end of the text.